Author's response to reviews

Title: Necrotic gangrenous intrathoracic appendix in a marfanoid adult patient: a case report

Authors:

Mohannad J Barakat (mbarakat@blueyonder.co.uk)
Jon H Vickers (jvickers@waht.swest.nhs.uk)

Version: 4 Date: 10 February 2005

Author's response to reviews:

I have revised the format of the paper so that the title is one page long, the affiliations are in the journal's format, the abstract is divided into background, case report and conclusions, and the rest of the manuscript is in the order of abstract, background, case report, conclusions, abbreviations, competing interests, acknowledgments and references.